Author’s response to reviews

Title: Maternal Safety of the Delayed-Release Doxylamine and Pyridoxine Combination for Nausea and Vomiting of Pregnancy; A Randomized Placebo Controlled Trial

Authors:

Gideon Koren (pharmtox@sickkids.ca)
Shannon Clark (shclark@utmb.edu)
Gary D.V. Hankins (ghankins@utmb.edu)
Steven N. Caritis (scaritis@mail.magee.edu)
Jason G. Umans (jgu@georgetown.edu)
Menachem Miodovnik (menachem.miodovnik@medstar.net)
Donald R. Mattison (dm425@nih.gov)
Ilan Matok (ilanmatok@gmail.com)

Version: 6 Date: 27 January 2015

Author’s response to reviews:

Dear Editor,

We included a Conclusions section in the manuscript and uploaded a revised manuscript.

Keywords were added as well as e-mail addresses for each author on cover page.

The TRN was put at the end of the abstract.
Thank you for all your help.